Catastrophic bleaching risks to Mesoamerican coral reefs in recent climate change projections

Meridel Murphy Phillips¹, Manishka De Mel², Anastasia Romanou³, David Rind⁴, Alex C Ruane⁵, and Cynthia Rosenzweig³

¹SciSpace, LLC
²Columbia University
³NASA Goddard Institute for Space Studies
⁴Goddard Institute for Space Studies
⁵Climate Impacts Group, NASA GISS, USA

November 26, 2022

Abstract

The damaging effects of climate change and increased ocean temperatures are already visible in marine ecosystems worldwide. Degree heating weeks (DHW) provide a valuable metric for gauging excess sea surface temperature warming and coral bleaching risk. This study produces future DHW projections for the Mesoamerican reef (MAR) using a multi-model climate change ensemble. We show that current marine heat wave conditions linked to coral bleaching will be far exceeded in an average year by mid-century, creating an environment where MAR corals have no opportunity for normal year recovery between extreme years. The dramatic increase in DHW in the MAR indicates strong adaptation interventions need to be developed and implemented as soon as possible to support local communities in adjusting to the effects of a warming climate.

Hosted file

essoar.10511564.1.docx available at https://authorea.com/users/554997/articles/605849catastrophic-bleaching-risks-to-mesoamerican-coral-reefs-in-recent-climate-changeprojections

Hosted file

dhwinmar_supportinginfo.docx available at https://authorea.com/users/554997/articles/605849catastrophic-bleaching-risks-to-mesoamerican-coral-reefs-in-recent-climate-changeprojections M. Phillips^{1,2}, M. De Mel^{3,2}, A. Romanou², D. Rind², A. C. Ruane², and C. Rosenzweig²

¹SciSpace LLC, Bethesda, Maryland, USA.

²NASA Goddard Institute for Space Studies, New York, New York, USA.

³Columbia University, New York, New York, USA.

Corresponding author: Meridel Phillips (mmp2192@columbia.edu)

Key Points:

- Projected increases in sea surface temperatures lead to high risk of permanent coral damage in the Mesoamerican reef
- Cumulative and consecutive marine heat stress events in the Mesoamerican reef allow low chance for coral recovery in climate projections

Abstract

The damaging effects of climate change and increased ocean temperatures are already visible in marine ecosystems worldwide. Degree heating weeks (DHW) provide a valuable metric for gauging excess sea surface temperature warming and coral bleaching risk. This study produces future DHW projections for the Mesoamerican reef (MAR) using a multi-model climate change ensemble. We show that current marine heat wave conditions linked to coral bleaching will be far exceeded in an average year by mid-century, creating an environment where MAR corals have no opportunity for normal year recovery between extreme years. The dramatic increase in DHW in the MAR indicates strong adaptation interventions need to be developed and implemented as soon as possible to support local communities in adjusting to the effects of a warming climate.

Plain Language Summary

Coral reefs are increasingly vulnerable to rising ocean temperatures, which can cause coral bleaching and mortality if sustained for long periods of time. Climate models can project the amount of marine warming and the risk to coral reefs using metrics that have been linked to historical bleaching events. We use the Degree Heating Week (DHW) indicator to show the potential for coral bleaching in the Mesoamerican reef under future climate change scenarios. An ensemble of 10 climate models shows that environmental conditions which are currently linked to severe bleaching will be exceeded in the average year by mid-century, creating a situation where corals have no opportunity for recovery between years with extremely high ocean temperatures. These results indicate that strong climate change adaptation interventions need to be developed and implemented as soon as possible to support local communities in the Mesoamerican reef.

1 Introduction

The Mesoamerican reef system (MAR), spanning more than 1,000 kilometers along the coastlines of Mexico, Belize, Honduras, and Guatemala, is the largest such expanse in the Northern Hemisphere and includes the world's second longest barrier reef (Gress et al., 2019; McField et al., 2008). The MAR is widely considered to be a region of global importance for biodiversity, containing more than 60 species of coral and 500 species of fish (De Mel et al., 2021), as well as a large network of habitats such as seagrasses, lagoons, and mangroves (Wilkinson & Souter, 2008). These coastal systems are critical for many of the region's economic activities, such as fisheries, tourism, and recreation, sustaining the livelihoods of more than two million people (De Mel et al., 2021; Green et al., 2017).

The MAR protects the coastal regions from erosion and damage from extreme weather events by reducing the intensity of waves and flooding, which additionally enables the formation of critical marine wildlife habitats and the protection of local economies (Hoegh-Guldberg, 1999). While the MAR helps support coastal communities against the negative impacts of climate change, the reef ecosystems are vulnerable to its escalating effects. These impacts include coral bleaching from warmer ocean temperatures (Baker et al., 2008; Brown, 1997; De Mel et al., 2021; Donner, 2011; Eakin et al., 2010; Eddy et al., 2021; Frieler et al., 2013; Goreau et al., 1992; Glynn & D'Croz, 1990; Hoegh-Guldberg, 1999; Langlais et al., 2017; McField et al., 2008; McWilliams et al., 2005; Nielsen et al., 2018; Sully et al., 2019; Teneva et al., 2012; Wilkinson & Souter, 2008), erosion and flooding of coastal areas due to extreme storms and sea level rise (Beck et al., 2018; Cuttler et al., 2018; Reguero et al., 2021), and other changing ocean parameters such as acidification (Anthony et al., 2008). The MAR also underwent amongst the largest coral bleaching events globally over the last twenty years (Sully et al., 2019). Figure 1 shows historical sea surface temperature (SST) anomalies across the broader Caribbean region for four recorded bleaching events in 1995, 1998, 2005 and 2015 using satellite data from the National Oceanic and Atmospheric Administration (NOAA) Coral Reef Watch (CRW), a global early-warning system that tracks conditions linked to coral bleaching (Liu et al., 2006).



Figure 1. Sea surface temperature anomalies in the MAR during historical bleaching episodes. NOAA mean SST anomalies during recorded bleaching events in the MAR for (a) October 1 – November 30, 1995, (b) September 18 – October 1, 1998, (c) July 15 – November 15, 2005, and (d) October 1 – November 30, 2015.

This work was motivated by Climate-Smarting Marine Protected Areas and Coastal Management in the Mesoamerican Reef Region, an initiative to accelerate the implementation of climate-smart tools in coastal communities and local governments (De Mel et al., 2021). The project is a collaboration between the World Wildlife Fund, Stanford University and Columbia University in partnership with government agencies and implementing partners from all four countries. Climate risk information on mean and extreme temperatures, precipitation levels, sea level rise and sea surface temperatures were developed in collaboration with input from local stakeholders with the purpose of integration into decision-making practices to guide shoreline management (De Mel et al., 2021). The results presented here are meant to further inform not only these objectives but also future climate impacts and adaptation applications.

While global projections of the climate-change-induced threats to corals already exist, little is known about the future of the reefs specifically in the MAR (Beyer et al., 2018). The purpose of this research is to provide a large ensemble of multi-model, multi-scenario, and multi-generational experiments to show how SST warming in the MAR will impact the risk of coral bleaching and mortality in the coming decades.

2 Climate and Coral Bleaching

2.1 Bleaching Thresholds

Coral reefs are economically valuable and rich in species diversity but are vulnerable to detrimental impacts from human activities such as pollution, overfishing and the broad effects of climate change (Anthony et al., 2008; Baker et al., 2008; Beck et al., 2018; Brown, 1997; Cramer et al., 2021; Cuttler et al. 2018; De Mel et al., 2021; Donner, 2011; Eakin et al., 2010; Eddy et al., 2021; Frieler et al., 2013; Goreau et al., 1992; Glynn & D'Croz, 1990; Hoegh-Guldberg, 1999; Knowlton & Jackson, 2008; Langlais et al., 2017; McField et al., 2008; McWilliams et al., 2005; Nielsen et al., 2018; Sully et al., 2019; Teneva et al., 2012; Wilkinson & Souter, 2008). Frequent bleaching events are already threatening the existence of coral reefs on a global scale, as reef systems have begun to show the long-term effects of their exposure to marine heat waves (Fox-Kemper et al., 2021; Frieler et al., 2013; Lough et al., 2018). Biodiversity in coral reefs is estimated to have declined by more than 60% since the 1950s as the extent of living coral has declined by half worldwide (Eddy et al., 2021). Even under the United Nations Framework Convention on Climate Change Paris Agreement target of 1.5°C warming above pre-industrial conditions, climate model projections of thermal stress on sea surface temperatures indicate a high likelihood of more frequent mass bleaching events (Lough et al., 2018). This intensifying ocean warming is likely to affect the survival of many coastal ecosystems and their protection of shoreline economies (Fox-Kemper et al., 2021).

Coral bleaching occurs when the relationship between coral and endosymbiotic algae is stressed by external factors, resulting in the coral discarding their algae (Brown, 1997; Nielsen et al., 2018). If this occurs over a prolonged period without the re-establishment of coral-algal symbiosis, the result is coral mortality (Brown, 1997). Figure 2 shows an example photograph of coral after experiencing bleaching conditions.



Figure 2. Example photograph of bleached coral. Bleached coral becomes white in appearance due to the loss of endosymbiotic algae. Photo provided by

Dr. Rich Aronson.

Although coral bleaching has been primarily linked to warmer ocean temperatures, coral health can be impacted by many other factors, including ocean currents, salinity, wind speed, hurricanes, sedimentation, and disease (Anthony et al., 2008; Baumann et al., 2019; Cuttler et al., 2018; Grimsditch & Salm, 2006; Hoegh-Guldberg, 1999; Knowlton & Jackson, 2008; McField et al., 2008; Nielsen et al., 2018; Wilkinson & Souter, 2008). Mass coral bleaching, however, is generally associated with higher-than-average ocean temperatures and solar radiation conditions (Eakin et al., 2010), with concurrent or compounding events contributing to vulnerable bleaching conditions and amplifying their effects (Dzwonkowski et al., 2020). Over time, repeated and chronic contributing factors can diminish reef resilience and have long-term effects on reef system health (Carilli et al., 2009; Grimsditch & Salm, 2006; Hughes et al., 2010). However, temperature-induced mass coral bleaching is only one of many threats facing the MAR (Ranasinghe et al., 2021). Increasing atmospheric CO_2 and declining ocean pH levels may result in lower levels of coral calcification and subsequent loss (Anthony et al., 2008); anthropogenic nutrient pollution conditions such as high inorganic nitrogen combined with low phosphate levels might expedite bleaching (Wiedenmann et al., 2013), and sea level rise will provide additional challenges for local communities of people in the fishing and tourism industries who rely on the reefs for their livelihoods (Perry et al., 2018). Coral bleaching can serve as an indicator for overall reef health, which provides critical information on the outlook for biodiversity and coastal economies over the coming decades.

The concept of a static temperature-based "bleaching threshold" varies, as different coral species are vulnerable at ocean temperatures that are anomalously higher than their species-specific baseline (Glynn & D'Croz, 1990; Teneva et al., 2012). Thus, it is important to assess past bleaching events to understand some of the underlying environmental conditions and compounding factors that lead to coral damage as well as management interventions that can help to avoid the worst mortality effects. Although smaller-scale bleaching events due to local stresses, extreme storms or increased temperatures were observed in the MAR before 1980, many of these events were followed by near-complete recovery (Glynn, 1984; Goreau et al., 1992; Wilkinson, 1999). Since that time, mass mortality events due to large-scale global temperature rise and concurrent weather extremes are becoming both more frequent and more severe as ocean temperatures increase (Glynn, 1984; Wilkinson, 1999).

2.2 Historical Bleaching Events in the MAR

2.2.1 1995 Bleaching Event

A widespread bleaching event occurred in the MAR in October and November of 1995 (Guzman & Guevara, 1998). Records indicate that ocean temperatures and solar radiation levels were anomalously high and wind speeds somewhat low (McField, 1999). Effects of bleaching varied widely by coral species, with one survey estimating the highest single-species mortality rate at 74% (Guzman & Guevara, 1998). Reefs across the Yucatán Peninsula as well as Belize and Honduras were severely impacted, although corals in the Western Caribbean escaped the worst effects (Hughes et al., 2018). The 1995 event also illustrates the connectivity between bleaching and longer-term effects on reef health, as it may have been associated with the subsequent outbreak of Black Band Disease (Guzman & Guevara, 1998).

2.2.2 1998 Bleaching Event

More significant damage was recorded after the 1998 bleaching event that occurred between September and October, which was followed by Hurricane Mitch in October-November 1998. Higher-than-average SSTs were recorded in the Yucatán Peninsula in Mexico in August 1998, spreading to Belize and Honduras in September (Kramer & Kramer, 2000). Category 5 Hurricane Mitch occurred in late October of that year, causing additional damage from severe winds and flooding (Kramer & Kramer, 2000). At the time of the hurricane, bleaching mortality had already occurred across much of the region, resulting in most of the storm damage impacting reef structures through physical devastation from storm waves and excessive sediment (Kramer & Kramer, 2000). Live coral cover mortality rates reached 50% through a combination of temperature and storm effects (McField et al., 2008).

2.2.3 2005 Bleaching Event

Another bleaching event in 2005 caused considerable damage, although there had been little regrowth between 1999 and 2005 (McField et al., 2008). Between June and November of 2005, satellite data showed SSTs at anomalously high levels compared to long-term averages not only in the MAR, but across the tropical Atlantic Ocean, with regional average SSTs at their warmest levels in the preceding 150 years (Eakin et al., 2010). Although widespread bleaching was recorded after the 2005 event, particularly in the Western Caribbean, coral mortality rates in the MAR were lower than those from the 1995 and 1998 events (Wilkinson & Souter, 2008). This was partially due to reduced coral cover of vulnerable species after the destruction of the 1998 event, as well as improved management decisions between 1998 and 2005 involving the reduction of local stressors such as pollution, overfishing, and overuse in tourism and recreational activities (McField et al., 2008; Schuttenberg & Marshall, 2008).

2.2.4 More Recent Bleaching Episodes

Moderate bleaching was reported near Belize in 2015, with more severe coral damage towards Cuba, the Cayman Islands, and the northern Caribbean region (Hughes et al., 2018). Coral core analysis of several sites on the coast of Belize indicated high thermal stress in near-shore reefs (Baumann et al., 2019). However, some studies indicate coral bleaching in 2015 was less severe than in previous events (Muñiz-Castillo & Arias-González, 2021). More frequent high-temperature events may have led to some reef resilience as the species survivors are becoming more tolerant to thermal stress, although at the cost of reef biodi-

versity as heat-intolerant species are unable to recover and may be permanently lost under continually warming conditions (Muñiz-Castillo & Arias-González, 2021).

3 Data and Methods

3.1 Degree Heating Weeks (DHW)

Recently, studies have shown the value of degree heating weeks (DHW) as a metric to indicate the severity of heat stress and potential for coral bleaching (Kayanne, 2017; Kayanne et al., 2017, Leggat et al., 2022). Although approaches have defined marine heat waves in slightly different ways (Hobday et al., 2016), the DHW approach used here is consistent with the NOAA CRW (Liu et al., 2006) and is among frequently used metrics in marine heat wave studies (Hobday et al., 2016; Kayanne, 2017; Kayanne et al., 2017, Leggat et al., 2022). Unlike static ocean temperature thresholds, the DHW indicator identifies cumulative heat stress measured by "HotSpots" of higher-than-average SST anomalies over a 12-week period (Liu et al., 2006), which have been linked to the timing and severity of coral bleaching and mortality not only in the MAR but across the globe (Eakin et al., 2010; Muñiz-Castillo & Arias-González, 2021). Significant coral bleaching generally happens when DHW exceed 4°C-weeks, with severe bleaching and likely significant mortality occurring at 8°C-weeks (Eakin et al., 2016).

During the 2005 bleaching event in the MAR, satellite and field data confirmed the accuracy of NOAA's DHW indicator in predicting the intensity of coral bleaching (see Figure S3a), with DHW reaching up to 16°C-weeks in some areas (Eakin et al., 2010). Additional studies across sites in the northwestern Pacific Ocean connect mass bleaching events to periods when the maximum DHW exceeded 8°C-weeks (Kayanne, 2017). Although the definition of DHW is not region- or species-specific and relies only on sea surface temperature inputs, it consistently correlates well with spatial and temporal patterns of bleaching worldwide (Donner, 2011; Heron et al., 2016; Langlais et al., 2017).

Previous studies have focused on correlating extreme or cumulative SST anomalies with recorded bleaching in reef systems across the globe (Eakin et al., 2010; Glynn & D'Croz, 1990; McWilliams et al., 2005). Several studies have used climate models to highlight the importance of concurrent stressors in regionspecific settings, or the contribution of seasonal variability in the level of vulnerability (Langlais et al., 2017; Melbourne-Thomas et al., 2011). Here we focus on the magnitude of DHW in future scenarios relative to an observational baseline for the MAR, the frequency of weekly SST anomalies that contribute to highvalue DHW, and the maximum consecutive duration of those anomalies, all of which can be considered as different indicators of the likelihood, severity, and length of marine heat waves.

DHW are calculated using SST anomalies from the climatological maximum mean month in a 7-year period from 1985-1993, excluding the years 1991 and 1992 due to the Mt. Pinatubo eruption (Liu et al., 2006). Monthly mean SST

values are calculated over this 7-year period, with the maximum mean month (MMM_SST_CLIM) from this period as the baseline from which SST anomalies are calculated. Fig. 3a shows the value of MMM_SST_CLIM across the region using observational data from MERRA-2 reanalysis (described below). Regional averages in this study apply only to the area of interest outlined in black in Fig. 3. Although coral reefs are generally close to the coast, we use a larger area to avoid mixed coastal effects in the climate models and to capture broader warming effects.

As DHW are calculated twice-weekly, HotSpots are determined for each half-week in the historical period by subtracting the MMM_SST_CLIM from mean SSTs in that half-week and removing negative values. DHW are calculated for each half-week using a sum of HotSpots over the 12 preceding weeks, only for those HotSpots greater than 1 degree (Equation 1).

$$DHW = 0.5 * \sum_{i=1}^{i=24} [\text{HotSpots}_i > 1C]_{\text{halfweekly}}$$

(Eqn 1)

The resulting DHW can be produced for each half-week for a given location. To facilitate the communication of DHW results, we refer to "annual DHW" throughout this study as the multi-year average across annual maximum DHW values over any specified period.

3.2 Historical and future data sources

This analysis uses the Modern-era Retrospective analysis for Research and Applications, Version 2 (MERRA-2) as a historical baseline (Gelaro et al., 2017). MERRA-2 is an assimilated dataset utilizing the Goddard Earth Observing System (GEOS) model along with satellite and in-situ observations and provided by NASA's Global Modeling and Assimilation Office (GMAO). For climate simulations, we analyze 10 Global Climate Models (GCMs) spanning two generations of Coupled Model Intercomparison Projects (CMIP5 and CMIP6), with three models from CMIP5 (Taylor et al., 2012) and seven from the newer-generation CMIP6 (Evring et al., 2016), listed in Table 1. As an example within our ensemble of GCMs, we highlight the GISS E2.1-G model, a state-of-the-art ocean model that is fully coupled to the atmosphere, land, and sea-ice components (Kelley et al., 2020). The ocean component has 1x1.25-degree horizontal resolution with 40 vertical layers, with higher resolution near the surface of about 5-10m, and natural surface boundary conditions for heat and freshwater fluxes (Kelley et al., 2020; Miller et al., 2021; Nazarenko et al., 2022). Table 1 also includes global warming levels (GWLs) for each GCM and scenario for a midcentury period, calculated from a pre-industrial reference period (1850-1900). Common global warming levels have become increasingly useful indicators for comparing GCMs, as the level of warming is suggestive of associated earth system effects and global impacts (Ranasinghe et al., 2021). Global warming

levels are calculated using area-weighted global mean temperatures to identify the temporal "crossing point" of a common set of global temperature anomalies from the reference period (Hauser et al., 2019). To calculate DHW in future scenarios, we use a "delta" approach by applying monthly SST warming amounts from GCMs for each future scenario to observational SSTs from MERRA-2 (see Supplementary Information for detailed materials and methods, including other bias-adjustment methods).

We analyze data from nine future decades of generated SST data using two different anthropogenic emission scenarios that are more comparable between CMIP5 and CMIP6. These scenarios are obtained from the Representative Concentration Scenarios (RCPs) or Shared Socioeconomic Pathways (SSPs) from CMIP5 or CMIP6 respectively. We include RCP4.5/SSP2-4.5 and RCP8.5/SSP5-8.5, which represent a "middle of the road" socioeconomic pathway with a nominal 4.5W/m² radiative forcing level by 2100, and a very high emissions pathway with a forcing of 8.5W/m², respectively (IPCC, 2021; Meinshausen et al., 2019; O'Neill et al., 2016). We focus on a mid-century period of 2050-2059, centered on the year 2054. GWLs for 2054 are interpolated from a list of 20-year windows associated with the available GWLs from each GCM. In addition to these GWLs derived from Hauser et al. (2019), we provide the area-averaged SST warming for each GCM across the rectangular region outlined in Figs. 3 and 5, calculated using SST data with a 1980-2009 reference period.

Table 1. GCMs included in multi-model ensemble. GCMs from CMIP5 and CMIP6 included in this study, with global warming level (GWL) from 1850-1900 and regional SST warming in the MAR from 1980-2009 for two emissions scenarios for the 2050s (both in °C). GWLs for GCMs, ensemble members and scenarios are derived from Hauser et al. (2019). Each GCM's ensemble member is also included as an "ripf" identifier, which categorizes the model's realization, initialization, physics and forcing indices (Taylor et al., 2018).

| GCM | SSP2-4.5 | SSP5-8.5 |
|--------------|---------------------------------------|--|
| | GWL at 2054 (°C), 1850-1900 reference | MAR SST change for 2050s (°C), 1980-2009 r |
| ACCESS1-0 | >2‡ | 1.29 |
| ACCESS-CM2 | 2.45 | 1.73 |
| CanESM5 | 3.15 | 1.76 |
| GFDL-CM4 | >2 | 1.48 |
| GISS-E2-R | >1.5 | 0.92 |
| GISS E2.1-G | 1.98 | 1.37 |
| IPSL-CM6A-LR | 2.66 | 1.31 |
| MRI-ESM2-0 | >2 | 1.36 |
| NorESM1-M | 1.72 | 0.9 |
| NorESM2-LM | 1.48 | 1.33 |

† Ensemble member from each model as denoted in the PCMIDI-AR6 database.

 $\ddagger > symbols indicate limited GWL data availability; see Table S1.$

4 Results

4.1 Projected Changes in DHW

For this analysis we select a mid-21st century period (2050-2059) to show the effects of warming over the next 30 years. Figs. 3b and 3c show the ensemble mean change in annual DHW between baseline and future for the 2050-2059 period in both emissions scenarios. Broad increases in DHW are visible across the outlined MAR region, well above the known bleaching and mortality thresholds of 4°C-weeks and 8°C-weeks. Increases in DHW in the MAR are slightly lower than in the southwest Atlantic in both scenarios, but are still dramatic, indicating that most future years will include significantly anomalously high SST values with weeks above dangerous DHW thresholds, even by mid-century.



Figure 3. Climatological maximum mean month and ensemble mean changes in annual DHW. (a) MERRA-2 climatological maximum mean monthly SST (MMM_SST_CLIM) from 1985-1993, excluding 1991-2; this represents the baseline from which HotSpots are calculated. Ensemble mean changes in annual DHW across ensemble of GCMs for 2050-2059 period, calculated from 1980-2009 baseline, are shown for (b) SSP2-4.5 scenario and (c) SSP5-8.5 scenario.

Figure 4 shows the change in annual DHW as a regional average across the boxed area outlined in Figure 3. We show mean decadal DHW for the ensemble of 10 GCMs as well as the MERRA-2 baseline. Shaded areas represent the range between lowest and highest projected mean annual DHW for each decade, with the mean across models shown as a bold line. We extend the model "envelope" to the last decadal mean in MERRA-2, although model decadal points are centered on the middle-year (e.g., 2014 for the 2010-2019 decade).



Figure 4. Regional decadal DHW from baseline to 2100. Time series of decadal DHW for MERRA-2 and future generated SST data using all GCMs and both scenarios. Shaded areas represent the range of lowest and highest projected mean annual DHW for each scenario and decade with ensemble means shown as bold lines.

The dramatic rise in annual DHW in Fig. 3 is again visible, with much more drastic impacts in the high-emissions scenario (SSP5-8.5). As these smoothed time series show only decadal averages, these do not include projections of extreme DHW correlating to marine heat wave events. The coastal ecosystem, biodiversity, and socioeconomic impacts of high-value DHW may therefore be even more dire than the values seen here, with current bleaching conditions becoming near-constant by mid-century. Even with smoothed data, the ensemble projects annual DHW in the region to increase by 7.9°C-weeks – 20.8°C-weeks under the lower-emissions scenario by the 2050s, with that range expanding to 13.2°C-weeks – 30.6°C-weeks under SSP5-8.5. By the 2090s, the upward trend in SSP2-4.5 projections has slowed due to the stabilization of greenhouse gas emissions in RCP4.5, but the range in projected increases is wider (8.9°C-weeks on the low end and 30.6°C-weeks on the high end). In the worst-case scenario, by the end of the century under SSP5-8.5, annual DHW may increase by 25.3°C-weeks – 55.6°C-weeks.

We additionally analyze metrics on the frequency and duration of HotSpots (SST anomalies greater than 1°C), values that are accumulated in each halfweekly DHW. As DHW are calculated using a rolling 12-week window, single half-weekly HotSpots may appear in multiple DHW. This approach is critical in defining DHW as enduring heat stress, but when assessing length and duration of heat spells, it is important to use non-recurring SST anomalies. For presentation, we apply a factor of 0.5 to half-weekly results to show the frequency and duration metrics in weeks per year. Fig. 5 shows the change in mean annual frequency of weekly HotSpots for the 2050-2059 period under both emissions scenarios, as well as the change in the maximum length of consecutive weekly HotSpots for the same period.



Figure 5. Ensemble mean changes in annual frequency and maximum consecutive duration of weekly HotSpots. Ensemble mean changes in annual frequency of weekly HotSpots for 2050-2059 period, calculated from 1980-2009 baseline, for (a) SSP2-4.5 scenario and (b) SSP5-8.5 scenario, and ensemble mean changes in annual maximum duration of consecutive weekly HotSpots for 2050-2059 period, calculated from 1980-2009 baseline, for (c) SSP2-4.5 scenario and (d) SSP5-8.5 scenario.

Without the cumulative effect of DHW calculation, spatial variability is more noticeable in both frequency and duration changes across the broader Caribbean region than in annual DHW changes (Fig. 3). Given significantly larger annual DHW in the future, the frequency of HotSpots is projected to increase substantially due to high levels of warming in regional SSTs. Relative spatial patterns in the changes of the maximum duration of warm spells (Figs. 5c, 5d) roughly match those seen in changes in frequency (Figs. 5a, 5b), with the highest values further south near Panama and along the southern coast of Guatemala and El Salvador. This indicates that the highest increases in the frequency of extreme events may correspond to much longer consecutive periods with SST anomalies above 1°C. The length of these warm spells, and the resulting extreme increases in DHW in both scenarios, imply severe consequences for the health of the MAR. Longer-length marine heat waves will likely be connected to higher incidences of coral bleaching and mortality, and increases in frequency suggest that the MAR may not have time to recover from consistently high SSTs in between events.

These changes in annual DHW may mask variability in the seasonal cycle of the average year in both baseline and future scenarios; therefore, it is useful to analyze the seasonal occurrence of high DHW values in comparison to present conditions. An average year of half-weekly DHW from the MERRA-2 baseline (1980-2009) is shown in black in Fig. 6 along with the spread of GCM projections of seasonal DHW in the 2050s under both scenarios, as in Fig. 4.



Figure 6. Average seasonal cycle of half-weekly DHW in baseline and future conditions. Average seasonal cycle of half-weekly DHW for baseline MERRA-2 (black) and range of GCMs under SSP2-4.5 (orange) and SSP5-8.5 (red). Shaded areas represent the full range of lowest and highest projected mean annual DHW for each scenario with ensemble means shown as bold lines. MERRA-2 seasonal cycle of DHW is shown in higher detail as an inset on the right.

The seasonal pattern of DHW visible in MERRA-2, with the lowest DHW in March and April after the cold months, is preserved in future scenarios, although this may be largely influenced by the delta method of bias-adjustment. The annual cycle of DHW peaks in October and November in both baseline and future scenarios, but the magnitude of future DHW greatly exceeds those seen in MERRA-2 in every month of the year. The highest DHW values in the MERRA-2 average are exceeded by nearly all GCM projections in all months of the year, except for the lowest estimates in the coolest months.

4.2 Data limitations and dimensions of uncertainty

This analysis uses SST data from combined GCM sources, providing a unique multi-model, multi-scenario perspective with a regional focus. However, additional factors could add new insights beyond the results presented here. First, we focus exclusively on the moderate (SSP2-4.5) and very high (SSP5-8.5) emission scenarios, with no results from the more optimistic SSP1-2.6. This adds a dimension of temporal uncertainty, with lower emissions and more mitigation measures in SSP1-2.6 allowing for more time before the crossing of critical warming thresholds. The range of GCM SST projections, even within this ensemble, provides another element of uncertainty; reducing inter-model spread would greatly benefit areas with strong adaptation needs such as the MAR. Additionally, these results use sea surface temperature data only, without the inclusion of vertical ocean profiles from individual GCMs. The incorporation of temperatures at different ocean depths can be important when studying the effects of warming on various marine species; however, we focus on SST data here due to the often shallow-water habitats of most coral species and the broad availability of observational and GCM SST outputs. Finally, at 0.25-degree spatial resolution, we interpolate GCM data to an extent where a much higher spatial variability is visible than in raw GCM ocean variables. Still higher-resolution SST products from global climate models with improved ocean processes and more complex model downscaling methods as well as advanced satellite data, such as that available from the NASA PACE Early Adopters Program (Cetinić et al., 2022), will contribute to improving projections for location-based planning and policymaking. Although the drastic warming results and the potential damages that we see from this ensemble are unlikely to be significantly altered with the addition of other temporal, vertical or spatial dimensions, these could provide new insights for those studying the effects of climate change with specific marine species, time horizons, emissions pathways, or mitigation scenarios.

5 Conclusions

Although specific to the MAR, this work provides insight into the link between climate change and the risk of significant coral bleaching. Despite spatial variability and GCM uncertainty in the magnitude of warming across the globe, warmer temperatures in many regions are all but guaranteed by mid-century under even more optimistic mitigation scenarios (Fox-Kemper et al., 2021; IPCC, 2021). As SSTs rise, the frequency, severity, duration, and spatial extent of marine heat waves also increases, a phenomenon which has already been observed in recent decades (Fox-Kemper et al., 2021; Pörtner et al., 2019). The level of warming during different parts of the seasonal cycle may vary, but even with moderate warming in cooler months, the traditionally "warm season" in the MAR will likely extend to include most of the year. When compared to current conditions, these changes indicate that the occurrence of consecutive weeks above critical DHW thresholds will be nearly inevitable. The definition of DHW as a metric that calculates SST anomalies relative to a region's climatological baseline gives it a portability that traditional thresholds lack, as the magnitude of difference between current and future climate conditions is a main contributor in calculating coral bleaching risk. This suggests that at a certain point, near-permanent warm season marine heat waves will make surface corals unviable in many regions, as the modes of variability in GCM projections likely present a larger source of uncertainty than regional disparities (Fox-Kemper et al., 2021; IPCC, 2021; Pörtner et al., 2019; Ranasinghe et al., 2021).

Despite rapid warming in the MAR, research indicates that coastal ecosystems can develop a level of natural resistance after exposure to warmer sea temperatures, although this often serves only as a temporary safeguard in the context of long-term warming (Munday et al., 2008). Studies that incorporate natural coral adaptation to higher SSTs show significantly reduced risk of bleaching across future scenarios (Frieler et al., 2013; Langlais et al., 2017; Logan et al., 2014; Teneva et al., 2012). However, compared to other regions, the MAR has historically seen less recovery after marine heat waves, due to a combination of factors such as the level and frequency of episodes, the affected species, and other concurrent extreme events (Baker et al., 2008). In research on the global consequences of warming on reef systems, the Caribbean is consistently classified among the most vulnerable, with thermal protection projected to decline significantly even under the ambitious target of 1.5°C above pre-industrial conditions (Dixon et al., 2022) and substantial declines in coral cover evident even in simulations of evolutionary adaptation responses (McManus et al., 2021).

Consequently, reef survival in the MAR depends on more than limiting emissions, although global mitigation of greenhouse gas emissions is still critical to delay the onset and reduce the severity of future marine heat waves to allow for the possibility of reef recovery. Reef restoration studies indicate moderate potential in adaptation methods such as coastal zone and fisheries management, structural restoration or physical reconstruction, and pollution reduction (Kleypas et al., 2021). More direct measures such as coral gardening through transplantation or micro-fragmentation, larval propagation, and artificial reef construction also show the capacity for coral regrowth in short-term research (Boström-Einarsson et al., 2020). Still, the success of adaptation measures depends strongly on the complexities of local governance, technological and financial limitations, and the genetic makeup of regional reef systems (Kleypas et al., 2021), as well as ongoing, long-term, and region-specific research on the efficacy of coral regrowth interventions (Boström-Einarsson et al., 2020).

Acknowledgments

MP, MDM, ACR, and CR were supported by the NASA Earth Science Division funding for the GISS Climate Impacts Group. The authors acknowledge Maria Dombrov for assistance with the preparation of figures for publication, as well as Dr. Rich Aronson for providing coral photography (Fig. 2). This research also made use of computing resources and support from the NASA Center for Climate Simulation (NCCS).

Open Research

Raw data and software: Raw data from NOAA are available from the NOAA Coral Reef Watch website (https://coralreefwatch.noaa.gov/index.php). Raw data from MERRA-2 are available from MDISC, managed by the NASA Goddard Earth Sciences (GES) Data and Information Services Center (DISC; https://disc.gsfc.nasa.gov/datasets?project=MERRA-2). Raw GCM data from CMIP5 and CMIP6 are available from various nodes on ESGF (https://esgf.llnl.gov/). This research made use of MATLAB and R software for analysis (MATLAB, 2018; R Core Team, 2018).

Availability statement: To make all data and analysis behind this study publicly available, we provide the regridded (0.25-degree) daily SST data from each of 10 GCMs as well as MERRA-2 in an online Goddard Institute for Space Studies (GISS) Climate Impacts data portal maintained by the NASA Center for Climate Simulation (NCCS, 2022). All of the figures in this manuscript and supporting information, with the exception of Figures 1 and S3a, can be recreating using the data on the GISS Impacts portal. Figures 1 and S3a can be recreated using data directly available from the NOAA website above. Table 1 is derived from data on the GISS Impacts portal as well as data from the Hauser et al. (2019) dataset. Table S1 provides the relevant data directly from Hauser et al. (2019) which can be used to interpolate values in Table 1. We additionally provide scripts for processing and analyzing the daily SST data from GCMs and MERRA-2, which are available on GitHub (Phillips, 2022). These scripts use MATLAB version 9.5.0.1049112 (R2018b) Update 3, and R version 3.5.2 (MATLAB, 2018; R Core Team, 2018). Both the GISS Impacts data portal and the GitHub repository cited here are accessible to the public with no registration required.

References

Anthony, K. R., Kline, D. I., Diaz-Pulido, G., Dove, S., & Hoegh-Guldberg, O. (2008). Ocean acidification causes bleaching and productivity loss in coral reef builders. *Proceedings of the National Academy of Sciences*, 105(45), 17442-17446. https://doi.org/10.1073/pnas.0804478105

Baker, A. C., Glynn, P. W., & Riegl, B. (2008). Climate change and coral reef bleaching: An ecological assessment of long-term impacts, recovery trends and future outlook. *Estuarine, Coastal and Shelf Science, 80*(4), 435-471. https://doi.org/10.1016/j.ecss.2008.09.003

Baumann, J. H., Ries, J. B., Rippe, J. P., Courtney, T. A., Aichelman, H. E., Westfield, I., & Castillo, K. D. (2019). Nearshore coral growth declining on the Mesoamerican Barrier Reef System. *Global Change Biology*, 25(11), 3932-3945. https://doi.org/10.1111/gcb.14784

Beck, M. W., Losada, I. J., Menéndez, P., Reguero, B. G., Díaz-Simal, P., & Fernández, F. (2018). The global flood protection savings provided by coral reefs. *Nature Communications*, 9(1), 1-9. https://doi.org/10.1038/s41467-018-04568-z

Bentsen, M., Bethke, I., Debernard, J. B., Iversen, T., Kirkevåg, A., Seland, Ø., et al. (2013). The Norwegian Earth System Model, NorESM1-M–Part 1: description and basic evaluation of the physical climate. *Geoscientific Model Development*, 6(3), 687-720. https://doi.org/10.5194/gmd-6-687-2013

Beyer, H. L., Kennedy, E. V., Beger, M., Chen, C. A., Cinner, J. E., Darling, E. S., et al. (2018). Risk-sensitive planning for conserving coral reefs under rapid climate change. *Conservation Letters*, 11(6), e12587. https://doi.org/10.1111/conl.12587

Bi, D., Dix, M., Marsland, S. J., O'Farrell, S., Rashid, H., Uotila, P., et al. (2013). The ACCESS coupled model: description, control climate and evaluation. *Australian Meteorological and Oceanographic Journal*, 63(1), 41-64. https://doi.org/10.22499/2.6301.004

Bi, D., Dix, M., Marsland, S. J., O'Farrell, S., Sullivan, A., Bodman, R., et al. (2020). Configuration and spin-up of ACCESS-CM2, the new generation Australian Community Climate and Earth System Simulator Coupled Model. *Journal of Southern Hemisphere Earth Systems Science*, 70(1), 225-251. https://doi.org/10.1071/ES19040

Boström-Einarsson, L., Babcock, R. C., Bayraktarov, E., Ceccarelli, D., Cook, N., Ferse, S. C., et al. (2020). Coral restoration–A systematic review of current methods, successes, failures and future directions. *PloS One*, 15(1), e0226631. https://doi.org/10.1371/journal.pone.0226631

Boucher, O., Servonnat, J., Albright, A. L., Aumont, O., Balkanski, Y., Bastrikov, V., et al. (2020). Presentation and evaluation of the IPSL-CM6A-LR climate model. *Journal of Advances in Modeling Earth Systems*, 12(7), e2019MS002010. https://doi.org/10.1029/2019MS002010

Brown, B. E. (1997). Coral bleaching: causes and consequences. *Coral Reefs*, 16(1), S129-S138. https://doi.org/10.1007/s003380050249

Carilli, J. E., Norris, R. D., Black, B. A., Walsh, S. M., & McField, M. (2009). Local stressors reduce coral resilience to bleaching. *Plos One*, 4(7), e6324. https://doi.org/10.1371/journal.pone.0006324

Cetinić, I., McClain, C. R., Werdell, P. J., & Behrenfeld, M. (2022). *PACE Technical Report Series, Volume 10: ACE Ocean Product Accuracy Assessments: A record of the state of the art circa 2010* (NASA/TM-2022-219027/Vol. 10). Greenbelt, MD: NASA ACE Ocean Field and Laboratory Measurement Group.

Cramer, K. L., Bernard, M. L., Bernat, I., Gutierrez, L., Murphy, E. L., Sangolquí, P., et al. (2021). The present and future status of ecosystem services for coral reefs. *Reference Module in Earth Systems and Environmental Sciences*. Elsevier. https://doi.org/10.1016/B978-0-12-821139-7.00177-X.

Cuttler, M. V. W., Hansen, J. E., Lowe, R. J., & Drost, E. J. F. (2018). Response of a fringing reef coastline to the direct impact of

a tropical cyclone. Limnology and Oceanography Letters, 3(2), 31-38. https://doi.org/10.1002/lol2.10067

De Mel, M., Phillips, M., Bartlett, R., Porta, M. A., Calzada, A., Bood., N., et al. (2021). *Climate Risk Information for the Mesoamerican Reef Region*. New York, NY: Center for Climate Systems Research at Columbia University, WWF-US and WWF-Mesoamerica.

Dixon, A. M., Forster, P. M., Heron, S. F., Stoner, A. M., & Beger, M. (2022). Future loss of local-scale thermal refugia in coral reef ecosystems. *PLoS Climate*, 1(2), e0000004. https://doi.org/10.1371/journal.pclm.0000004

Donner, S. D. (2011). An evaluation of the effect of recent temperature variability on the prediction of coral bleaching events. *Ecological Applications*, 21(5), 1718-1730. https://doi.org/10.1890/10-0107.1

Dunne, J. P., Horowitz, L. W., Adcroft, A. J., Ginoux, P., Held, I. M., John, J. G., et al. (2020). The GFDL Earth System Model version 4.1 (GFDL-ESM 4.1): Overall coupled model description and simulation characteristics. *Journal of Advances in Modeling Earth Systems*, 12(11), e2019MS002015. https://doi.org/10.1029/2019MS002015

Dzwonkowski, B., Coogan, J., Fournier, S., Lockridge, G., Park, K., & Lee, T. (2020). Compounding impact of severe weather events fuels marine heatwave in the coastal ocean. *Nature Communications*, 11(1), 1-10. https://doi.org/10.1038/s41467-020-18339-2

Eakin, C. M., Morgan, J. A., Heron, S. F., Smith, T. B., Liu, G., Alvarez-Filip, L., et al. (2010). Caribbean corals in crisis: record thermal stress, bleaching, and mortality in 2005. *PloS One*, 5(11), e13969. https://doi.org/10.1371/journal.pone.0013969

Eddy, T. D., Lam, V. W., Reygondeau, G., Cisneros-Montemayor, A. M., Greer, K., Palomares, M. L. D., et al. (2021). Global decline in capacity of coral reefs to provide ecosystem services. *One Earth*, 4(9), 1278-1285. https://doi.org/10.1016/j.oneear.2021.08.016

Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937-1958. https://doi.org/10.5194/gmd-9-1937-2016

Fox-Kemper, B., Hewitt, H. T., Xiao, C., Aðalgeirsdóttir, G., Drijfhout, S. S., Edwards, T. L., et al. (2021). Ocean, Cryosphere and Sea Level Change. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1211–1362). Cambridge, UK and New York, NY: Cambridge University Press. https://doi.org/10.1017/9781009157896.011 Frieler, K., Meinshausen, M., Golly, A., Mengel, M., Lebek, K., Donner, S. D., & Hoegh-Guldberg, O. (2013). Limiting global warming to 2C is unlikely to save most coral reefs. *Nature Climate Change*, 3(2), 165-170. https://doi.org/10.1038/nclimate1674

Garcia, D. (2021). Inpaint over missing data in 1-D, 2-D, 3-D,... ND arrays [Soft-ware]. MATLAB Central File Exchange. https://www.mathworks.com/matlabcentral/fileexchange/27994-inpaint-over-missing-data-in-1-d-2-d-3-d-nd-arrays.

Gelaro, R., McCarty, W., Suárez, M. J., Todling, R., Molod, A., Takacs, L., et al. (2017). The modern-era retrospective analysis for research and applications, version 2 (MERRA-2). *Journal of Climate*, 30(14), 5419-5454. https://doi.org/10.1175/JCLI-D-16-0758.1

Glynn, P. W. (1984). Widespread coral mortality and the 1982-83 El Niño warming event. Environmental Conservation, 11(2), 133-146. https://doi.org/10.1017/S0376892900013825

Glynn, P. W., & D'Croz, L. (1990). Experimental evidence for high temperature stress as the cause of El Niño-coincident coral mortality. *Coral Reefs*, 8(4), 181-191. https://doi.org/10.1007/BF00265009

Goreau, T. J., Hayes, R. L., Clark, J. W., Basta, D. J., & Robertson, C. N. (1992). Elevated sea surface temperatures correlate with Caribbean coral reef bleaching. In R. A. Geyer (Ed.), *Global Warming Forum: Scientific, Economic, and Legal Overview* (pp 225-255). Boca Raton, FL: CRC Press.

Green, A., Chollett, I., Suárez, A., Dahlgren, C., Cruz, S., Zepeda, C., et al. (2017). Biophysical Principles for Designing a Network of Replenishment Zones for the Mesoamerican Reef System. Mexico: The Nature Conservancy, Comunidad y Biodiversidad, A.C., Smithsonian Institution, Perry Institute for Marine Science, Centro de Estudios Marinos, Healthy Reefs Initiative and Universidad Autónoma de Baja California Sur. https://doi.org/10.13140/RG.2.2.18562.79044

Gress, E., Voss, J. D., Eckert, R. J., Rowlands, G., & Andradi-Brown, D. A. (2019). The Mesoamerican Reef. In Y. Loya, K. Puglise, T. Bridge (Eds.), *Mesophotic Coral Ecosystems* (Vol. 12, pp. 71-84). Switzerland: Springer.

Grimsditch, G. D., & Salm, R. V. (2006). Coral Reef Resilience and Resistance to Bleaching. Gland, Switzerland: IUCN.

Guzman, H. M., & Guevara, C. (1998). Massive mortality of zooxanthelate reef organisms during the 1995 bleaching in Cayos Cochinos, Honduras. *Revista de Biologia Tropical*, 46, 165-173.

Hauser, M., Engelbrecht, F., & Fischer, E. M. (2019). Transient global warming levels for CMIP5 and CMIP6 (v0.1.0) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.3591807

Heron, S. F., Johnston, L., Liu, G., Geiger, E. F., Maynard, J. A., De

La Cour, J. L., et al. (2016). Validation of reef-scale thermal stress satellite products for coral bleaching monitoring. *Remote Sensing*, $\delta(1)$, 59. https://doi.org/10.3390/rs8010059

Hobday, A. J., Alexander, L. V., Perkins, S. E., Smale, D. A., Straub, S. C., Oliver, E. C., et al. (2016). A hierarchical approach to defining marine heatwaves. *Progress in Oceanography*, 141, 227-238. https://doi.org/10.1016/j.pocean.2015.12.014

Hoegh-Guldberg, O. (1999). Climate change, coral bleaching and the future of the world's coral reefs. *Marine and Freshwater Research*, 50(8), 839-866. https://doi.org/10.1071/MF99078

Hughes, T. P., Anderson, K. D., Connolly, S. R., Heron, S. F., Kerry, J. T., Lough, J. M., et al. (2018). Spatial and temporal patterns of mass bleaching of corals in the Anthropocene. *Science*, 359(6371), 80-83. https://doi.org/10.1126/science.aan8048

Hughes, T. P., Graham, N. A., Jackson, J. B., Mumby, P. J., & Steneck, R. S. (2010). Rising to the challenge of sustaining coral reef resilience. *Trends in Ecology & Evolution*, 25(11), 633-642. https://doi.org/10.1016/j.tree.2010.07.011

IPCC. (2021). Summary for Policymakers. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), *Climate Change* 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 3-32). Cambridge, UK and New York, NY: Cambridge University Press. https://doi.org/10.1017/9781009157896.001

Iversen, T., Bentsen, M., Bethke, I., Debernard, J. B., Kirkevåg, A., Seland, \emptyset ., et al. (2013). The Norwegian earth system model, NorESM1-M–Part 2: climate response and scenario projections. *Geoscientific Model Development*, 6(2), 389-415. https://doi.org/10.5194/gmd-6-389-2013

Kayanne, H. (2017). Validation of degree heating weeks as a coral bleaching index in the northwestern Pacific. Coral Reefs, 36(1), 63-70. https://doi.org/10.1007/s00338-016-1524-y

Kayanne, H., Suzuki, R., & Liu, G. (2017). Bleaching in the Ryukyu Islands in 2016 and associated degree heating week threshold. *Galaxea, Journal of Coral Reef Studies*, 19(1), 17-18. https://doi.org/10.3755/galaxea.19.1_17

Kelley, M., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Ruedy, R., Russell, G. L., et al. (2020). GISS-E2. 1: Configurations and climatology. *Journal of Advances in Modeling Earth Systems*, 12(8), e2019MS002025. https://doi.org/10.1029/2019MS002025

Kleypas, J., Allemand, D., Anthony, K., Baker, A. C., Beck, M. W., Hale, L. Z., et al. (2021). Designing a blueprint for coral reef survival. *Biological Conservation*, 257, 109107. https://doi.org/10.1016/j.biocon.2021.109107

Knowlton, N., & Jackson, J. B. C. (2008). Shifting baselines, local

impacts, and global change on coral reefs. *PLoS Biology*, 6(2), e54. https://doi.org/10.1371/journal.pbio.0060054

Kramer, P. A., & Kramer, P. R. (2002). Transient and lethal effects of the 1998 coral bleaching event on the Mesoamerican reef system. In *Proceedings* of the 9th International Coral Reef Symposium (Vol. 2, pp. 1175-1180). Bali, Indonesia.

Langlais, C. E., Lenton, A., Heron, S. F., Evenhuis, C., Sen Gupta, A., Brown, J. N., & Kuchinke, M. (2017). Coral bleaching pathways under the control of regional temperature variability. *Nature Climate Change*, 7(11), 839-844. https://doi.org/10.1038/nclimate3399

Leggat, W., Heron, S. F., Fordyce, A., Suggett, D. J., & Ainsworth, T. D. (2022). Experiment Degree Heating Week (eDHW) as a novel metric to reconcile and validate past and future global coral bleaching studies. *Journal of Environmental Management*, 301, 113919. https://doi.org/10.1016/j.jenvman.2021.113919

Liu, G., Strong, A. E., Skirving, W., & Arzayus, L. F. (2006). Overview of NOAA coral reef watch program's near-real time satellite global coral bleaching monitoring activities. In *Proceedings of the 10th International Coral Reef Symposium* (pp. 1783-1793). Okinawa, Japan.

Logan, C. A., Dunne, J. P., Eakin, C. M., & Donner, S. D. (2014). Incorporating adaptive responses into future projections of coral bleaching. *Global Change Biology*, 20(1), 125-139. https://doi.org/10.1111/gcb.12390

Lough, J. M., Anderson, K. D., & Hughes, T. P. (2018). Increasing thermal stress for tropical coral reefs: 1871–2017. Scientific Reports, $\delta(1)$, 1-8. https://doi.org/10.1038/s41598-018-24530-9

MATLAB. (2018). version 9.5.0.1049112 (R2018b) Update 3 [Software]. Natick, Massachusetts: The MathWorks Inc.

McField, M. D. (1999). Coral response during and after mass bleaching in Belize. *Bulletin of Marine Science*, 64(1), 155-172.

McField, M., Bood, N., Fonseca, A., Arrivillaga, A., Rinos, A. F., & Viruel, R. M. L. (2008). Status of the Mesoamerican Reef after the 2005 coral bleaching event. In C. Wilkinson, D. Souter (Eds.), *Status of Caribbean coral reefs after bleaching and hurricanes in 2005* (pp. 45-60). Townsville, Australia: Global Coral Reef Monitoring Network, and Reef and Rainforest Research Centre.

McManus, L. C., Forrest, D. L., Tekwa, E. W., Schindler, D. E., Colton, M. A., Webster, M. M., et al. (2021). Evolution and connectivity influence the persistence and recovery of coral reefs under climate change in the Caribbean, Southwest Pacific, and Coral Triangle. *Global Change Biology*, 27(18), 4307-4321. https://doi.org/10.1111/gcb.15725

McWilliams, J. P., Côté, I. M., Gill, J. A., Sutherland, W. J., & Watkinson, A. R. (2005). Accelerating impacts of temperature-induced coral bleaching in the

Caribbean. Ecology, 86(8), 2055-2060. https://doi.org/10.1890/04-1657

Meinshausen, M., Smith, S. J., Calvin, K., Daniel, J. S., Kainuma, M. L., Lamarque, J. F., et al. (2011). The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. *Climatic Change*, 109(1), 213-241. https://doi.org/10.1007/s10584-011-0156-z

Melbourne-Thomas, J., Johnson, C. R., & Fulton, E. A. (2011). Regionalscale scenario analysis for the Meso-American Reef system: Modelling coral reef futures under multiple stressors. *Ecological Modelling*, 222(10), 1756-1770. https://doi.org/10.1016/j.ecolmodel.2011.03.008

Miller, R. L., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Kelley, M., Ruedy, R., et al. (2021). CMIP6 Historical Simulations (1850–2014) With GISS-E2.
1. Journal of Advances in Modeling Earth Systems, 13(1), e2019MS002034. https://doi.org/10.1029/2019MS002034

Munday, P. L., Jones, G. P., Pratchett, M. S., & Williams, A. J. (2008). Climate change and the future for coral reef fishes. *Fish and Fisheries*, 9(3), 261-285. https://doi.org/10.1111/j.1467-2979.2008.00281.x

Muñiz-Castillo, A. I., & Arias-González, J. E. (2021). Drivers of coral bleaching in a Marine Protected Area of the Southern Gulf of Mexico during the 2015 event. *Marine Pollution Bulletin*, 166, 112256. https://doi.org/10.1016/j.marpolbul.2021.112256

Nazarenko, L. S., Tausnev, N., Russell, G. L., Rind, D., Miller, R. L., Schmidt, G. A., et al. (2022). Future Climate Change under SSP Emission Scenarios with GISS-E2. 1. *Journal of Advances in Modeling Earth Systems*, e2021MS002871. https://doi.org/10.1029/2021MS002871

Nielsen, D. A., Petrou, K., & Gates, R. D. (2018). Coral bleaching from a single cell perspective. The ISME Journal, 12(6), 1558-1567. https://doi.org/10.1038/s41396-018-0080-6

NCCS. (2022). Dataportal - Datashare: GISS/Impacts/Ecosystems/MesoamericanReef/DailySSTs/ [Dataset]. NASA Center for Climate Simulation. https://portal.nccs.nasa.gov/datashare/GISS/Impacts/Ecosy

O'Neill, B. C., Tebaldi, C., Van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9(9), 3461-3482. https://doi.org/10.5194/gmd-9-3461-2016

Perry, C. T., Alvarez-Filip, L., Graham, N. A., Mumby, P. J., Wilson, S. K., Kench, P. S., et al. (2018). Loss of coral reef growth capacity to track future increases in sea level. *Nature*, 558(7710), 396-400. https://doi.org/10.1038/s41586-018-0194-z

Phillips, M. (2022). mmp2192/MesoamericanReef-DHW: v1.0: Catastrophic Bleaching Risks to Mesoamerican Coral Reefs in Recent Climate Change Projections [Software]. https://doi.org/10.5281/zenodo.6585274

Pörtner, H.-O., Roberts, D. C., Masson-Delmotte, V., Zhai, P., Tignor, M., Poloczanska, E., et al. (2019). *IPCC Special Report on the Ocean and Cryosphere in a Changing Climate*. Cambridge, UK and New York, NY: Cambridge University Press. https://doi.org/10.1017/9781009157964

R Core Team. (2018). R: A language and environment for statistical computing. Vienna, Austria: R Foundation for Statistical Computing (Version 3.5.2) [Software]. https://www.R-project.org/

Ranasinghe, R., Ruane, A. C., Vautard, R., Arnell, N., Coppola, E., Cruz, F. A., et al. (2021). Climate Change Information for Regional Impact and for Risk Assessment. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1767–1926). Cambridge, UK and New York, NY: Cambridge University Press. https://doi.org/10.1017/9781009157896.014

Reguero, B. G., Storlazzi, C. D., Gibbs, A. E., Shope, J. B., Cole, A. D., Cumming, K. A., & Beck, M. W. (2021). The value of US coral reefs for flood risk reduction. *Nature Sustainability*, 4(8), 688-698. https://doi.org/10.1038/s41893-021-00706-6

Schmidt, G. A., Kelley, M., Nazarenko, L., Ruedy, R., Russell, G. L., Aleinov, I., et al. (2014). Configuration and assessment of the GISS ModelE2 contributions to the CMIP5 archive. *Journal of Advances in Modeling Earth Systems*, 6(1), 141-184. https://doi.org/10.1002/2013MS000265

Schuttenberg, H., & Marshall, P. (2008). Managing for mass coral bleaching: Strategies for supporting socio-ecological resilience. In C. Wilkinson, D. Souter (Eds.), *Status of Caribbean coral reefs after bleaching and hurricanes in 2005* (pp. 115-128). Townsville, Australia: Global Coral Reef Monitoring Network, and Reef and Rainforest Research Centre.

Seland, Ø., Bentsen, M., Olivié, D., Toniazzo, T., Gjermundsen, A., Graff, L. S., et al. (2020). Overview of the Norwegian Earth System Model (NorESM2) and key climate response of CMIP6 DECK, historical, and scenario simulations. *Geoscientific Model Development*, 13(12), 6165-6200. https://doi.org/10.5194/gmd-13-6165-2020

Sully, S., Burkepile, D. E., Donovan, M. K., Hodgson, G., & Van Woesik, R. (2019). A global analysis of coral bleaching over the past two decades. *Nature Communications*, 10(1), 1-5. https://doi.org/10.1038/s41467-019-09238-2

Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., et al. (2019). The Canadian earth system model version 5 (CanESM5. 0.3). *Geoscientific Model Development*, 12(11), 4823-4873. https://doi.org/10.5194/gmd-12-4823-2019

Taylor, K. E., Juckes, M., Balaji, V., Cinquini, L., Denvil, S., Durack, P., et al.

(2018). CMIP6 Global Attributes, DRS, Filenames, Directory Structure, and CV's [Dataset]. PCMDI Document. https://goo.gl/v1drZl

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. *Bulletin of the American Meteorological Society*, 93(4), 485-498. https://doi.org/10.1175/BAMS-D-11-00094.1

Teneva, L., Karnauskas, M., Logan, C. A., Bianucci, L., Currie, J. C., & Kleypas, J. A. (2012). Predicting coral bleaching hotspots: the role of regional variability in thermal stress and potential adaptation rates. *Coral Reefs*, 31(1), 1-12. https://doi.org/10.1007/s00338-011-0812-9

Wiedenmann, J., D'Angelo, C., Smith, E. G., Hunt, A. N., Legiret, F. E., Postle, A. D., & Achterberg, E. P. (2013). Nutrient enrichment can increase the susceptibility of reef corals to bleaching. *Nature Climate Change*, 3(2), 160-164. https://doi.org/10.1038/nclimate1661

Wilkinson, C. R. (1999). Global and local threats to coral reef functioning and existence: review and predictions. *Marine and Freshwater Research*, 50(8), 867-878. https://doi.org/10.1071/MF99121

Wilkinson, C. R., & Souter, D. (2008). Executive summary. In C. Wilkinson, D. Souter (Eds.), *Status of Caribbean coral reefs after bleaching and hurricanes in 2005* (pp. 3-14). Townsville, Australia: Global Coral Reef Monitoring Network, and Reef and Rainforest Research Centre.

Yukimoto, S., Kawai, H., Koshiro, T., Oshima, N., Yoshida, K., Urakawa, S., et al. (2019). The Meteorological Research Institute Earth System Model version 2.0, MRI-ESM2. 0: Description and basic evaluation of the physical component. Journal of the Meteorological Society of Japan. Ser. II, 97, 931-965. https://doi.org/10.2151/jmsj.2019-051